Author's response to reviews

Title: Promotion of the Mind Through Exercise (PROMoTE): A Proof-of-Concept Randomized Controlled Trial of Aerobic Exercise Training in Older Adults with Vascular Cognitive Impairment

Authors:

Teresa Liu-Ambrose (tlambrose@exchange.ubc.ca)
Janice J Eng (janice.eng@vch.ca)
Lara A Boyd (lara.boyd@ubc.ca)
Claudia Jacova (claudija@interchange.ubc.ca)
Jennifer C Davis (jcdavis@interchange.ubc.ca)
Stirling Bryan (stirling.bryan@ubc.ca)
Philip Lee (pelee@providencehealth.bc.ca)
Penny Brasher (penny.brasher@vch.ca)
Ging-Yuek R Hsiung (hsiung@interchange.ubc.ca)

Version: 2 Date: 10 February 2010

Author's response to reviews:

Dear Dr. Norton,

We have made all the necessary formatting changes to our manuscript. Please note our Trial Registration number is mentioned in our abstract. However, because our abstract is just over 2 pages, the Trial Registration number is on the second page.

Sincerely,

Teresa Liu-Ambrose